Author's response to reviews

Title: Significant Correlation between Urinary N1, N12-diacetylspermine and Tumor Invasiveness in Patients with Clinical Stage IA Non-small Cell Lung Cancer

Authors:

Yusuke Takahashi (yusuketakahashigts@gmail.com)
Hirotoshi Horio (hirohori@cick.jp)
Koji Sakaguchi (sakaguchi-kouji@pref-nagano-hosp.jp)
Kyoko Hiramatsu (hiramatsu-kk@igakuken.or.jp)
Masao Kawakita (kawakita-ms@igakuken.or.jp)

Version: 5
Date: 13 January 2015

Author's response to reviews: see over
Response to the Editor

Dear Ryan M. Relox,
Academic Editor, BMC CANCER

I enclosed the revised manuscript entitled “Significant Correlation between Urinary N$^{1}$, N$^{12}$-diacetylspermine and Tumor Invasiveness in Patients with Clinical Stage IA Non-small Cell Lung Cancer” (MS: 7929952551406451). We would be grateful if you could address the comments in a revised manuscript and provide a cover letter giving a point-by-point response to the concerns.

Yours sincerely,

Yusuke Takahashi, M.D., Ph.D.
Tokyo Metropolitan Cancer and Infectious Diseases Center Komagome Hospital, Tokyo, Japan

Response to Editor

Comment 1: We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

Response 1: The revised manuscript has already edited by Dr. Keith Kretzmer, Medical Communication Advisor of Department of General Thoracic Surgery of Teikyo University School of Medicine, who edited our original manuscript.

Comment 2: Please include funding details for your study and each author in your acknowledgements section.
Response 2: We have added the sentence, “This work was supported in part by a Grant-in-Aid for Scientific Research (C) from the Japan Society for the Promotion of Science (#21590639),” into the acknowledgements section.